PredictSURE-IBD and IBDX to guide personalised treatment of Crohn's disease in adults

DAR PROTOCOL

This report was commissioned by the NIHR HTA Programme as project number 128968



Title: PredictSURE-IBD and IBDX to guide personalised treatment of Crohn's disease in adults

Produced by:	BMJ Technology Assessment Group (BMJ-TAG)				
Authors:	Steven J Edwards, Director of Health Technology Assessment, BMJ-TAG, London				
	Samantha Barton, Principal Health Technology Assessment Analyst, BMJ-TAG, London				
	Mariana Bacelar, Principal Health Economist, BMJ-TAG, Londor				
	Charlotta Karner, Health Technology Assessment Analysis Manager, BMJ-TAG, London				
	Peter Cain, Senior Health Economist, BMJ-TAG, London				
	Shaji Sebastian, Consultant Gastroenterologist, IBD Unit, Department of Gastroenterology, Hull & East Yorkshire NHS Trust				
Correspondence to:	Steve Edwards, BMJ-TAG, BMJ, BMA House, Tavistock Square, London, WC1H 9JR.				
Date completed:	07/05/2019				

Source of funding: This protocol was commissioned by the NIHR HTA Programme as part of project number 128968.

Declared competing interests of the authors:

No competing interests were declared which affect the impartiality of this report. BMJ Technology Assessment Group (BMJ-TAG) and the editorial team of The BMJ work independently to one another. The views and opinions expressed in this report are those of the BMJ-TAG.

Acknowledgements:

The EAG would like to thank Professor Shaji Sebastian (Consultant Gastroenterologist) for providing clinical advice throughout the development of this protocol.

Rider on responsibility for protocol

The views expressed in this report are those of the protocol and not necessarily those of the NIHR HTA Programme. Any errors are the responsibility of the authors.

This report should be referenced as follows: Edwards SJ, Barton S, Bacelar M, Karner C, Cain P, Sebastian S. PredictSURE-IBD and IBDX to guide personalised treatment of Crohn's disease in adults – Protocol. BMJ Technology Assessment Group, 2019.

TABLE OF CONTENTS

Table of c	ontents	3				
Table of a	bbreviations	5				
1 PLAIN ENGLISH SUMMARY						
2 DE	DECISION PROBLEM					
2.1	Aim of the assessment	8				
2.2	Population and target condition	8				
2.2.1	Population: High risk of relapse associated with Crohn's disease	8				
2.2.2	Target condition: Crohn's disease	9				
2.2.3	Epidemiology	10				
2.2.4	Current diagnostic and treatment pathways	11				
2.	2.4.1 Diagnosis of Crohn's disease	11				
2.	2.4.2 Management of Crohn's disease	12				
2.3	Interventions	15				
2.4	Place of the interventions in the treatment pathway	18				
2.5	Relevant comparators					
2.6	Reference standard	-				
-	PORT METHODS FOR ASSESSING THE OUTCOMES ARISING FROM THE U					
TH	E INTERVENTIONS	19				
3.1	Review eligibility criteria	19				
3.1.1	Population	19				
3.1.2	Interventions	19				
3.1.3	Comparators	19				
3.1.4	Reference standard	19				
3.1.5	Outcomes	19				
3.1.6	Study design	21				
3.2	Search strategy	21				
3.3	Study selection	22				
3.4	Data extraction strategy					
3.5	Quality assessment strategy					
3.6	Methods of analysis/synthesis					
3.6.1	Potential subgroup analyses					
3.6.2	Sensitivity analyses					
4 REI	PORT METHODS FOR SYNTHESISING EVIDENCE OF COST EFFECTIVENES					
4.1	Identifying and systematically reviewing published cost-effectiveness studies					
4.2	Development of a health economic model					
4.2.1	Model structure					
	NDLING INFORMATION FROM THE COMPANIES					
6 CO	MPETING INTERESTS OF AUTHORS	29				

7	TIMETABLE/MILESTONES	.30
8	REFERENCES	.31
9	APPENDICES	.36
9.1	Appendix 1. Draft search strategy (MEDLINE)	.36

TABLE OF ABBREVIATIONS

Abbreviation	In full
CASP	Critical Appraisal Skills Programme
CDAI	Crohn's Disease Activity Index
CDSR	Cochrane Database of Systematic Reviews
CRD	Centre for Reviews and Dissemination
CRP	C-reactive protein
DARE	Database of Abstracts and Reviews of Effects
EAG	Evidence assessment group
FCP	Faecal calprotectin
НВІ	Harvey–Bradshaw Index
HSUV	Health state utility values
НТА	Health technology assessment
IBD	Inflammatory bowel disease
ICER	Incremental cost-effectiveness ratio
NHS	National Health Service
NHS EED	NHS Economic Evaluation Database
NICE	National Institute for Health and Care Excellence
PCR	Polymerase chain reaction
PSA	Probabilistic sensitivity analysis
QALY	Quality-adjusted life year
QUADAS-2	Quality Assessment of Diagnostic Accuracy Studies tool – 2
RCT	Randomised controlled trial
ROBINS-I	Risk Of Bias In Non-randomised Studies – of Interventions
RT	Reverse transcriptase
SLR	Systematic literature review
TNF	Tumour necrosis factor
UK	United Kingdom

1 PLAIN ENGLISH SUMMARY

Crohn's disease is a type of inflammatory bowel disease that can affect any part of the digestive tract, and the section and size of the inflamed part of the intestine vary from person to person. The areas of inflammation are often irregular, with sections of healthy intestine between affected parts. The cause of Crohn's disease is not known. Environmental factors, such as smoking and certain medications (e.g., antibiotics), genes, and the immune system are thought to have a part in the development of and course of Crohn's disease.

Symptoms of the disease include pain in the abdomen, severe diarrhoea, tiredness, and weight loss. Crohn's disease can also cause problems outside the gastrointestinal tract, which are known as extraintestinal manifestations. People with Crohn's disease might also develop arthritis, and skin and eye problems. In severe cases, Crohn's disease can lead to life-threatening complications. Symptoms of Crohn's disease can be different from person to person, and there is no one test to identify the condition so diagnosing the disease is difficult.

Although there is no cure for Crohn's disease at this time, treatments are available that can manage the symptoms, and, in some cases, can lead to someone having no symptoms for a long time (remission). Drugs are given to reduce the inflammation, with a corticosteroid usually the first treatment. If symptoms do not improve with a corticosteroid, the next option is a drug that blocks the response of the immune system (immunosuppressant), which can be given on its own or with a corticosteroid. Some immunosuppressants are known as biologics (e.g., infliximab), and, although these therapies are more effective at improving symptoms, they can have side effects and are often used when symptoms are severe and do not respond to other treatments.

Some people with Crohn's disease have symptoms (known as a flare) much more often than others, and, as a result, are at a higher risk of complications of their disease. Symptoms can vary from mild cramping and diarrhoea to severe abdominal pain or blockages of the bowel. Identifying who will have flares more often is difficult, because there is no test available that can accurately predict long-term disease course. Using clinical judgement and a person's history, clinicians try to identify a person as being at risk of having a severe course of disease and decide to treat these people earlier with biologics. However, because of the possibility of side effects, there are concerns about giving biological therapies to people if symptoms could be managed with corticosteroids and immunosuppressants. Various biomarkers have been identified that are linked with having a higher risk of severe course of Crohn's disease, and diagnostic tools have been developed to identify people who have the characteristic biomarkers. Predicting which people are at risk of long-term complications of their Crohn's disease could lead to improved clinical outcomes because clinicians might be able to personalise treatment.

The aim of this project is to review the clinical scientific evidence, and to assess the costs and benefits associated with the use of two prognostic tools — the PredictSURE-IBDTM and the IBDX — to identify those with Crohn's disease at a high risk of having a severe course of disease.

2 DECISION PROBLEM

2.1 Aim of the assessment

The aim of this diagnostic assessment review is to assess the test accuracy and clinical and costeffectiveness of two molecular diagnostic tools for inflammatory bowel disease (IBD) in identifying those at a high-risk of severe course of disease, with a restriction to Crohn's disease for this project. The tools assessed in the review reported here are PredictSURE-IBDTM and the Crohn's disease Prognosis Test (IBDX). At the time of writing, no validated test or algorithm is available to stratify people with Crohn's disease by risk of developing complications of disease. Presence of known risk factors for flare and for complications in Crohn's disease could influence the treating clinician's management of the condition, but consensus on use of risk factors to determine prognosis of disease has yet to be achieved and treatment can vary. The accuracy, clinical and cost-effectiveness of the diagnostic tools will be evaluated against standard clinical care in the National Health Service (NHS), based on input from clinical advisors, when assessing the likely course of Crohn's disease.

2.2 Population and target condition

2.2.1 Population: High risk of severe course of Crohn's disease

IBD is characterised by chronic inflammation of the gastrointestinal tract and primarily comprises two subtypes of disorder – Crohn's disease and ulcerative colitis.¹⁻³ The symptoms of Crohn's disease and ulcerative colitis are similar, and both types of IBD affect men and women equally.¹⁻³ Crohn's disease and ulcerative colitis are lifelong conditions for which there is no cure, the courses of which are characterised by recurring cycles of exacerbation (also referred to as flare) and remission. The key differences between the conditions are the section of the intestine affected, and the extent of inflammation. Ulcerative colitis is limited to the colon, whereas Crohn's disease can occur anywhere between the mouth and the anus. Additionally, whereas ulcerative colitis affects only the inner most lining of the colon, Crohn's disease can penetrate into all layers of the bowel wall.

Flares of IBD indicate a return to active disease and, potentially, symptoms for an individual. Several factors have been proposed as triggers for flare, including poor adherence to treatment, certain medications (e.g., antibiotics and non-steroidal anti-inflammatory drugs), infection, smoking, and emotional stress.^{4,5} As has been noted in other immune-mediated diseases, the course of Crohn's disease and ulcerative colitis varies widely among affected individuals, making it challenging to predict the severity or frequency of occurrence of flare.

Endoscopic assessment and biopsies provide data on level of disease activity in IBD but do not give an insight into factors for relapse and course of disease. Evaluating blood and stool-based biomarkers of inflammation, such as C-reactive protein (CRP) and faecal calprotectin (FCP), is less invasive than

endoscopy and such laboratory tests provide reproducible, quantitative results that, together with clinical assessment, can aid clinicians in the diagnosis and management of IBD. However, serum and faecal biomarkers have limited application in the prediction of the severity of the course of IBD in the longer term.⁶ With no validated prognostic marker available, clinicians and patients use their judgement and experience to determine an individual's risk of developing a severe course of disease, with the goal of reducing or preventing future exacerbations of disease.

2.2.2 Target condition: Crohn's disease

Crohn's disease can affect any segment of the gastrointestinal tract, but the most commonly affected areas are the end of the ileum (the last part of the small intestine) and the colon.⁷ Diseased segments are frequently separated by intervening areas of healthy bowel tissue.^{2, 8} The size of the inflamed area may be limited to a few centimetres, or could affect an extensive part of the bowel.^{2, 8} As well as affecting the lining of the gastrointestinal tract, Crohn's disease may also penetrate the wall of the bowel.

As Crohn's disease can affect any part of the gastrointestinal tract, and to different extents, there is considerable variation in the symptoms experienced at the individual level by people with the disease, making recognition and diagnosis difficult.^{2, 8} Moreover, symptoms and severity of disease can change over time. People with Crohn's disease most commonly present with:^{2, 8, 9}

- abdominal pain;
- altered bowel pattern, including diarrhoea (mucus, pus or blood may be mixed with the diarrhoea);
- tiredness and fatigue;
- loss of appetite and weight loss;
- anaemia.

Neither the underlying aetiology of Crohn's disease nor the factors that determine the course and prognosis of the disease are fully understood. Environmental factors (e.g., smoking), genes, and the immune system are thought to have a part in the development of and course of Crohn's disease.^{2, 8} Those who develop Crohn's disease that follows a non-severe course might achieve prolonged remission with no treatment. With medical treatment, and with or without surgery:¹⁰

- about 50% of patients will be in remission or have mild disease over the subsequent 5 years;
- 45% of those in remission will remain relapse-free over the next year;

- 35% will have one or two relapses;
- 11% will have chronically active disease.

For those in remission with Crohn's disease after treatment, relapse rates at 1, 2, 5, and 10 years are estimated at 20%, 40%, 67%, and 76%, respectively.¹¹ In contrast to a non-complicated course of Crohn's disease, cases characterised as following a severe course are likely to experience more frequent flares, and typically require early aggressive treatment strategies, including multiple treatment escalations and augmentation. People with severe forms of Crohn's disease are at high risk of complications of disease, including intestinal obstruction, fistulae and perianal disease, and progressive disability and need for surgery.^{2, 8, 9}

Crohn's disease can also cause problems outside the gastrointestinal tract, which are known as extraintestinal manifestations.⁹ Associated conditions typically occur during flare, but can also manifest during remission or before development of any signs of IBD. Conditions developing as a result of Crohn's disease include:⁹

- arthritis (more commonly of the large joints of the arms and legs, including the elbows, wrists, knees and ankles);
- skin problems, most commonly erythema nodosum;
- eye problems (episcleritis, scleritis and uveitis).

Crohn's disease is a debilitating disease, having a marked impact on physical and emotional health, as well as quality of life. Additionally, Crohn's disease is associated with high economic burden due to disability, loss of work productivity, surgery and hospitalisation.¹² In 2015, a UK study estimated the annual cost of care for a person with Crohn's disease to be £6,156 (£1,800 for those in remission compared with £10,513 for those experiencing relapse),¹³ which translates to a total annual cost of ~£700 million. Five years after onset, 15% to 20% of people are affected to some degree by their disease, and between 50% and 80% of people with Crohn's disease will eventually need surgery as a result of, for example, development of strictures, perforation of the bowel, or failure of drug therapy.¹⁴

2.2.3 Epidemiology

Crohn's disease can appear at any age, but more than 90% of people will develop symptoms before the age of 40.¹⁵ In the UK, it is estimated that Crohn's disease affects one in every 650 people⁹ and that there are at least 115,000 people with the condition.⁸ Incidence and prevalence of Crohn's disease have been rising since the mid-1970s, with highest rates observed in Northern Europe and North America.¹⁶

Incidence of Crohn's disease in the UK is reported to be about 8 per 100,000 people per year,^{17, 18} with an age–sex adjusted point prevalence of 144.8 per 100,000 people.¹⁸

2.2.4 Current diagnostic and treatment pathways

2.2.4.1 Diagnosis of Crohn's disease

The symptoms of Crohn's disease are common to various conditions, which makes diagnosis of the condition challenging. Furthermore, there is no single test to diagnose definitively Crohn's disease, and diagnosis is reached through a combination of clinical examination, laboratory tests, imaging assessment, and endoscopy.¹⁹

On diagnosis of Crohn's disease, guidelines recommend that subsequent investigations focus on assessing level of activity of disease, as well as risk of complications in the longer term.²⁰ Three key areas are assessed when determining the severity of Crohn's disease: impact of the disease on the individual (e.g., clinical symptoms, quality of life, fatigue, and disability); burden of disease (e.g., C-reactive protein, mucosal lesions, upper gastrointestinal involvement, and disease extent); and course of disease (e.g., structural damage, perianal disease, number of flares, and extraintestinal manifestations).²¹

Two clinical tools available to assess level of disease activity are the Crohn's Disease Activity Index (CDAI)²² and the Harvey–Bradshaw Index (HBI).²³ The HBI is a simple derivative of the CDAI, and the two tools are correlated, with a change in the CDAI of 100 points corresponding to a 3-point change in the HBI.²⁴ Clinical experts commented that, in clinical practice, their preference is for the HBI tool, as the CDAI is impractical for routine clinical assessment and its use is typically limited to clinical trials. Severity of disease activity is categorised as:²⁵

- clinical remission: CDAI score of ≤ 150 , which corresponds to a HBI score of ≤ 4 ;
- mild: CDAI score of 150–220, which corresponds to a HBI score of 4–8;
- moderate-severe: CDAI score of 220–450, which corresponds to a HBI score of ≥ 8 ;
- severe fulminant disease: CDAI score of \geq 450, which corresponds to a HBI score of \geq 15.

IBD activity and severity could be considered a continuum, and some people might not easily be categorised based on their symptoms. Moreover, the CDAI and HBI are based on subjective measures, and there is a move to using more objective parameters and the presence or absence of bowel destruction to assess severity.²⁰ Using patient-reported outcomes to assess disease activity in Crohn's disease is also becoming more common. Often used to guide treatment recommendations, the CDAI and HBI scores

represent status of activity at a point in time and do not account for long-term prognosis or course of disease.²⁶

As well as assessing clinical parameters, clinicians might use physical characteristics, symptoms and results from imaging and laboratory to gauge a person's risk of experiencing a severe course of disease, and identify those who could benefit most from early use of aggressive treatments (immunosuppressors and biological therapies).²⁵ Prognostic factors associated with a more complicated disease course include bowel damage, extraintestinal manifestations of disease, number of flares, need for glucocorticoids, and resultant hospitalisations.²⁶ Other risk factors for a worse course of disease include smoking, and fistula formation. Factors at diagnosis of disease that were found to be associated with a worse prognosis were young age (<40 years), presence of perianal disease, and initial need for corticosteroid treatment.²⁵ A person's Crohn's disease may be considered to be following a severe course if they have refractory or relapsing disease necessitating multiple treatment escalations (dose increases and/or add-on treatment), development of significant complications (e.g., irreversible penetrating or stricturing lesions), need for more than one surgery, need for hospitalisation, or a combination of the listed factors. There is no consensus or algorithm available outlining how to combine known risk factors to determine long-term course of disease, and estimation of prognosis is based on subjective clinical judgement.

Being able to better predict the course of Crohn's disease would help clinicians and people with the condition to decide on the most appropriate treatment to manage symptoms. Tools such as the PredictSURE-IBD[™] and IBDX could potentially help achieve the goal of personalising treatment in Crohn's disease.

2.2.4.2 Management of Crohn's disease

As noted earlier, there is no cure for Crohn's disease and, at the time of writing, the goal of treatment is to initially control or reduce symptoms to induce remission.¹² Once symptoms are under control, maintenance treatment might be given to prolong remission and prevent relapse.¹² Globally, there are two treatment algorithms followed in the management of Crohn's disease – the "step up" and "top down" approaches (Figure 1) – both of which involve several tiers of medication, and, as the names suggest, are the inverse of each other.²⁷



Figure 1. "Step up" versus "top down" treatment algorithms for Crohn's disease

Abbreviation: 5-ASA, 5-aminosalicylate. Note: In the treatment hierarchy, the more potent drug therapies are placed at the top of the pyramid.

Currently, the NICE guideline (NG129¹²) recommends a "step up" approach for the management of Crohn's disease (Figure 1), which involves starting treatment with the least clinically effective of the options available and, in cases where disease does not respond to treatment as desired, escalating therapy in stages, with the strategy determined by clinical judgement and patient preference. If an adequate response is not apparent in the expected time frame, the "step up" plan can be accelerated.¹⁹ The Evidence Assessment Group's (EAG's) clinical expert advised that, for those people judged likely to benefit from the "step up" approach, most clinicians would prefer to accelerate treatment rather than follow the conventional "step up" algorithm.

The "top down" approach is currently not recommended by NICE.¹² The strategy involves treatment earlier in the pathway with those therapies that are more clinically effective, but that are also potentially associated with a greater risk of adverse effects. Early use of biological therapies in a "top down" approach is thought to modify the course of Crohn's disease, to increase the possibility of mucosal healing (prevents structural damage of the bowel), and to be more effective than the "step up" approach at inducing and prolonging remission.²⁷

The focus of treatment in Crohn's disease is to induce and maintain remission, to improve patient symptoms and quality of life, with a move towards personalising treatment. The goal of achieving mucosal healing during treatment is gaining acceptance but is not yet part of standard care in the UK. Neither the "step up" nor "top down" approach is suitable for all people with Crohn's disease. Considering the risk–benefit profile of the "top down" approach, some clinicians could be reticent to expose those with mild activity of Crohn's disease at time of assessment, or those thought to be at low risk of experiencing a relapse, to the unnecessary risk of an adverse effect. Conversely, those assessed as potentially being at risk of having a severe course of disease are at risk of undertreatment if the step up approach is followed, with consequent prolonging of symptoms and of inadequate control of disease

activity, and the associated long-term risks. Another consideration is cost of treatment, with the "top down" approach typically more expensive compared with the "step up" approach.²⁸

Ability to easily stratify those with Crohn's disease by risk of course of disease could help identify the most appropriate treatment strategy for the patient.

2.2.4.2.1.1 "Step up" approach

NICE NG129¹² advises starting treatment with a corticosteroid (prednisolone, methylprednisolone or intravenous hydrocortisone) for those with a first presentation or a single inflammatory exacerbation of Crohn's disease in a 12-month period. Alternative treatments for those who cannot tolerate, or who are contraindicated to, the recommended corticosteroids, are another corticosteroid, budesonide, and 5-aminosalicylate, both of which are less effective, but might be associated with fewer adverse effects, than preferred initial treatment:¹⁴ clinical experts advised that 5-aminosalicylate is no longer used in clinical practice in the UK. Budesonide should not be considered for those presenting with severe disease activity or exacerbations.

Should disease not respond to initial treatment with a corticosteroid, a subsequent treatment option is addition of an immunosuppressant (azathioprine or mercaptopurine) to a conventional glucocorticosteroid or budesonide if:¹²

- there are 2 or more inflammatory exacerbations in a 12-month period; or
- the glucocorticosteroid dose cannot be tapered.

Alternatively, if it is thought that the person would be unable to tolerate mercaptopurine or azathioprine, addition of methotrexate could be considered.

For adults with severe active Crohn's disease whose disease has not responded to conventional therapy (including immunosuppressive and/or corticosteroid treatments), or who are intolerant of or have contraindications to conventional therapy, recommended treatment is escalation to a tumour necrosis factor (TNF)-alpha inhibitor (infliximab or adalimumab) given as a monotherapy or in combination with an immunosuppressant.¹²

For those with moderately to severely active Crohn's disease and who have failed treatment with a TNFalpha inhibitor (i.e., disease has responded inadequately or lost response to treatment) or who are intolerant to conventional therapies, other biologics, such as vedolizumab and ustekinumab, are additional treatment options.¹²

Options for maintenance of remission are:¹²

- azathioprine or mercaptopurine as monotherapy after corticosteroids (including budesonide) to induce remission and for those who have not previously received these drugs;
- methotrexate:
 - For those who required methotrexate to induce remission;
 - For those who tried but could not tolerate azathioprine or mercaptopurine for maintenance;
 - For those are contraindicated to azathioprine or mercaptopurine.
- continued treatment with biologic, if appropriate.

2.2.4.2.2 "Top-down" approach

Although the "top-down" approach is not recommended by NICE, clinicians in specialist centres might choose to offer the strategy as an option to those they consider to have a poor prognosis in terms of outcomes, for example, those with complex perianal disease, significant fistulising disease or those with multiple risk factors. No accepted treatment strategy is available for the "top down" approach, with disparity across studies in the definition of "aggressive" therapy.²⁸ Additionally, evidence in support of the effectiveness of the "top down" approach is inconsistent, with some studies finding a benefit of early treatment with biologics and some reporting no benefit.²⁸ Variation in results across studies could be related to differences in, for example, the definition of "early" intervention and of outcomes measured, population, and trial duration.

2.3 Interventions

2.3.1 PredictSURE-IBD™

PredictSURE-IBD[™] is proposed for use in adults (16 years or older) with IBD, including Crohn's disease, who have active disease and are not receiving concomitant steroids, immunomodulators or biological therapies. PredictSURE-IBD[™] could be particularly beneficial for people with:

- newly or recently diagnosed IBD;
- moderate or severe active IBD (people with mild disease are unlikely to have early aggressive treatment with biologics);
- disease that would not require early aggressive treatment with biologics ('top-down' approach) with current standard care in the NHS (e.g., people who do not have fistulising and/or complex perianal Crohn's disease, or have multiple risk factors).

PredictSURE-IBD[™] facilitates stratification of people with IBD into high and low risk of frequently relapsing course of disease through detection of a gene sequence associated with CD8+ T cell exhaustion. Gene expression profiling of peripheral blood CD8+ T cells identified a signature gene sequence that was associated with CD8+ T cell exhaustion,²⁹⁻³¹ a state that is reached through the stepwise and progressive loss of T-cell function and that inhibits the immune response.³² Level of expression of the genes indicating CD8+ T cell exhaustion was found to be linked with course of disease in multiple autoimmune diseases, including IBD.²⁹⁻³¹ People with a CD8+ T cell signature not associated with T cell exhaustion were shown to be at higher risk of a frequently relapsing disease course than those with the signature for T cell exhaustion.²⁹⁻³¹ Early identification of those at high risk of recurrent course of Crohn's disease may lead to improved clinical outcomes through facilitation of personalised treatment, particularly in those with newly diagnosed disease.

The PredictSURE-IBD[™] test determines the presence or absence of the signature gene sequence (15 target genes and 2 control genes) indicating CD8+ T cell exhaustion through *in vitro* quantitative reverse transcription-polymerase chain reaction (RT-qPCR) of messenger RNA (mRNA) isolated from a whole blood sample (2.5 ml). The blood sample must be taken by a trained professional and stored in a sample tube (PAXgene Blood RNA tube): the vessel for the blood sample is not supplied as a component of the PredictSURE-IBD[™] test kit and must be purchased separately. Isolation of mRNA and subsequent RT-qPCR is carried out in a centralised laboratory (Clinical Genetics Laboratory, Addenbrooke's Treatment Centre, Cambridge University Hospitals NHS Foundation Trust).

In RT-qPCR, because the starting genetic material is RNA rather than DNA, the first step in the process requires transcription of mRNA into complementary DNA (cDNA) using reverse transcriptase (RT). Next, the cDNA acts as the template for qPCR for DNA amplification. qPCR is carried out in a 384-well plate (16 x 24 wells). Given the requirements for quality control of the assay, a maximum of 4 samples can be analysed per plate. Each sample of cDNA is amplified in triplicate, which requires 12 rows of the plate. A quality control RNA (supplied as part of the PredictSURE-IBDTM kit and run in triplicate [3 rows]) and a no-RNA control (run singularly [1 row]) are tested with each batch of mRNA samples to validate the run. The centralised laboratory uses a Roche LightCycler 480/480 II platform, which is a standard platform, to carry out RT-PCR. Staff training to process the PredictSURE-IBDTM kits will not be required at the centralised laboratory as the site is already providing testing services as part of an ongoing study (PROFILE). If required, PredictImmune would support staff training at additional laboratories to facilitate expansion of testing, with training thought require 2–3 days at each centre (draft scope from NICE).

Results from RT-qPCR are fed into a proprietary algorithm that calculates a continuous risk score, and based on this score, patients are categorised as high- or low-risk of following a frequently relapsing

form of IBD. A confidence level associated with the result is also reported and presented as a percentage. Turnaround time for the test is 7-10 days.

2.3.2 Crohn's disease Prognosis Test

The Crohn's disease Prognosis Test (IBDX) developed by Glycominds uses serological markers to identify those at risk of a more complicated course of Crohn's disease (development of strictures and fistulae, and need for surgery).

The abnormal interaction of environmental, genetic and microbial factors with the immune system leads to the dysregulated immune response that is responsible for the intestinal inflammation typical of Crohn's disease. Those with Crohn's disease have an atypical immune response to the normal bacteria found in the gut that leads to the production of antibodies against microbial components.³³ Examples include anti-Saccharomyces cerevisiae antibodies (ASCA), antibodies against Pseudomonas-associated sequence I2 (anti-I2), and against the bacterial flagellin cBir1 (Anti-cBir1).³³

A set of biomarkers that has been reported to be highly specific for Crohn's disease, with potential predictive value for prediction of complicated course of disease, are the anti-glycan-antibodies.³⁴ Glycans are saccharides that can be attached to various biological molecules through an enzymatic process called glycosylation. Glycans are usually found on the exterior of cell walls, and they form the main components of the cell wall surface in many microbes, including fungi, yeast, and bacteria.³⁴ Anti-glycan antibodies comprise anti-Saccharomyces cerevisiae antibodies (gASCA), anti-mannobioside antibodies (AMCA), anti-laminaribioside antibodies (ALCA), anti-chitobioside antibodies (ACCA), anti-laminaribioside antibodies (ALCA).

The IBDX tool detects antibodies including:35

- ACCA;
- ALCA;
- AMCA;
- gASCA;
- Anti-L;
- Anti-C.

The IBDX antibodies are detected in patient serum or plasma by an indirect solid-phase enzyme-linked immunosorbent assay (ELISA). Each kit (sold individually) contains the relevant anti-glycan 96-well

microwell plate (12 X 8 well strips), ELISA reagents, negative control, positive control, and calibrators. The microwell plates, conjugates and controls are specific for each kit, but all other reagents are the same. All kits follow the same procedure (including incubation times), so they can easily be processed at the same time, if desired. For each biomarker, positivity is assessed based on the cut-off values presented in Table 1.

Those with Crohn's disease are considered to be at greater risk for disease complication (stricturing or penetrating) or surgery intervention if they are positive for two or more serological markers.

	gASCA lgG	ALCA IgG	ACCA lgA	AMCA IgG	anti-C IgA	anti-L IgA	
Negative	<45	<55	<80	<90	<45	<45	
Equivocal ^a	45–50	55–60	80–90	90–100	45–50	45–50 >50	
Positive	>50	>60	>90	>100	>50		
a Repetition of sample assay is recommended.						•	

Table 1. Cut-off values for individual IBDX ELISA kits

2.4 Place of the interventions in the treatment pathway

The proposed placement of PredictSURE-IBD[™] and IBDX in the treatment pathway is in the stratification of those with a diagnosis Crohn's disease as high versus low risk of severe course of disease, thereby facilitating personalised treatment.

2.5 Relevant comparators

As no validated tool or algorithm is available to determine course of Crohn's disease, the relevant comparator is standard clinical care in the NHS.

2.6 Reference standard

As there is no test or algorithm available to determine long-term course of disease or an individual's risk of developing severe course of disease, estimation of prognosis is based on subjective clinical judgement of presenting signs and symptoms, together with potential risk factors for severe course of disease. Thus, there is no reference standard for the tools under evaluation.

3 REPORT METHODS FOR ASSESSING THE OUTCOMES ARISING FROM THE USE OF THE INTERVENTIONS

A systematic literature review will be carried out to evaluate the clinical effectiveness, costeffectiveness and diagnostic test accuracy of the PredictSURE-IBDTM and Crohn's disease Prognosis Test (IBDX) tools for the identification of those at high risk versus at low risk of developing a severe course of Crohn's disease.

A review of the evidence for clinical effectiveness will be undertaken systematically following the general principles outlined in the Centre for Reviews and Dissemination (CRD) guidance for conducting reviews in healthcare,³⁶ NICE's Diagnostics Assessment Programme manual³⁷ and the Cochrane Handbook for Systematic Reviews of Diagnostic Test Accuracy.³⁸

3.1 Review eligibility criteria

The eligibility criteria required for inclusion of a study in the review of the clinical effectiveness evidence are described in the subsections that follow.

3.1.1 Population

Study population eligible for inclusion will be those with a confirmed diagnosis of active Crohn's disease, and a diagnosis of disease.

3.1.2 Interventions

The PredictSURE-IBD[™] and IBDX diagnostic tools.

3.1.3 Comparators

No studies will be excluded based on type of comparator or lack of comparator.

3.1.4 Reference standard

As there is no test or algorithm available to determine long-term course of disease or an individual's risk of experiencing severe course of disease, estimation of prognosis is based on subjective clinical judgement. Thus, there is no reference standard for the tools under evaluation.

3.1.5 Outcomes

Evidence permitting, the outcomes listed below will be considered:

- diagnostic accuracy (e.g. sensitivity and specificity, and/ or if raw data are available, the numbers of true positive, true negative, false positive and false negative test results for predicting course of disease);
- diagnostic yield (number of diagnoses of severe versus non-severe course of Crohn's disease);
- time to test result;
- number of test failures;
- number of inconclusive test results;
- percentage of people for whom early treatment with biologics was offered ('top-down') by subgroup of severe versus non-severe course of disease;
- rates and duration of response and remission by subgroup of severe versus non-severe course of disease;
- rates and duration of flare-ups and/or relapses by subgroup of severe versus non-severe course of disease;
- rates and duration of corticosteroid-free remission by subgroup of severe versus non-severe course of disease;
- cumulative corticosteroid exposure by subgroup of severe versus non-severe course of disease;
- measures of mucosal healing by subgroup of severe versus non-severe course of disease;
- rates of and time to treatment escalation by subgroup of severe versus non-severe course of disease;
- rates of and time to hospitalisation by subgroup of severe versus non-severe course of disease;
- rates of and time to surgical intervention by subgroup of severe versus non-severe course of disease;
- rates of and time to serious complication (e.g., obstruction, intestinal ulcers, fistula, anal fissure) by subgroup of severe versus non-severe course of disease;

- composite outcomes formed of hospitalisation, surgery or serious complication (obstruction, intestinal ulcers, fistula, anal fissure) by subgroup of severe versus non-severe course of disease;
- adverse effects of treatment;
- health-related quality of life by subgroup of severe versus non-severe course of disease.

3.1.6 Study design

The highest level of evidence for assessment of the clinical effectiveness of the prognostic tool that is the focus of the systematic review outlined in this protocol would be a randomised controlled trial (or systematic review of such studies) that allocated those with Crohn's disease to treatment guided by assessment using the tool or to treatment based on clinical judgement. However, based on scoping searches, and given that the interventions are prognostic tools, retrieval of relevant RCTs is unlikely. Thus, to ensure that all relevant studies are captured, no limit relating to study design will be applied, with the exception that studies must be carried out in humans. Studies analysing the clinical validity (the ability of the test to reliably and accurately identify the biomarkers of interest or determine the risk of developing severe versus non-severe course of Crohn's disease) or clinical utility (the ability of the test to improve measurable clinical outcomes, and its usefulness and added value to patient management) of the prognostic tool will be eligible for inclusion. Studies evaluating analytical validity will be included where applicable, where analytical validity denotes the ability of the tool to accurately and reliably measure the biomarker of interest as assessed using laboratory tests on samples that are representative of those with Crohn's disease. Studies not published in English language will be included if sufficient relevant data can be extracted from the full-text publication in non-English language, or from an English language abstract. Studies excluded on the basis of language will be listed separately. Non-peer-reviewed reports or abstracts will only be included if the data are presented in a succinct and accessible manner (e.g. a manuscript prepared for submission to a journal), if sufficient methodological details are reported to allow critical appraisal of the study quality, and if results are reported in sufficient detail.

3.2 Search strategy

Search terms will be a combination of Medical Subject Headings (MeSH) and free text terms for the population of Crohn's disease and for relevant biomarkers. Terms for the interventions of interest and relevant alternative terms are included in consideration of future updates but it is noted that, based on scoping searches, inclusion of intervention names retrieves zero relevant records and terms must be combined with "or" to avoid omission of known potentially relevant studies. A flow diagram illustrating

the flow of information through the systematic review process will be presented according to the PRISMA reporting guidelines.

The systematic literature search will comprise the following main elements:

- Searching of electronic bibliographic databases;
- Contact with clinical experts in the field;
- Review of the reference lists of retrieved papers.

Electronic databases will be searched from inception until the latest available version. The electronic databases that will be searched are:

- MEDLINE (draft search strategy provided in Appendix 9.1);
- EMBASE;
- The Cochrane Central Register of Controlled Trials (CENTRAL).

Clinical trial registers will also be searched to identify relevant ongoing clinical trials that when completed may have an impact on the results of this review. Registers to be searched include:

- WHO International Clinical Trials Registry Platform;
- ClinicalTrials.gov (http://clinicaltrials.gov/).

The website of the US Food and Drug Administration (FDA) will also be searched to identify unpublished data.

Relevant reviews and guidelines will be identified through consultation with clinical experts and searching the National Institute for Health and Care Excellence (NICE) website to identify additional potentially relevant studies.

Reference lists of included papers will be assessed. If necessary and if time allows, the authors of eligible studies will be contacted for further information (e.g. full text of citations listed ahead of print).

The abstracts from key conference proceedings from the past 2 years, of conferences identified in consultation with clinical experts, will be screened, where possible, for additional potentially relevant studies.

3.3 Study selection

The titles and abstracts of studies retrieved from the electronic database searches will be independently assessed for potential relevance by two reviewers according to the prespecified eligibility criteria. In cases in which consensus cannot be achieved, the full texts of potentially relevant studies will be ordered. Full-text copies of potentially relevant studies will be obtained and assessed independently by two reviewers for inclusion against the prespecified eligibility criteria. Any disagreements will be resolved by discussion, or, a third reviewer will be consulted, if necessary.

3.4 Data extraction strategy

Data will be extracted by one reviewer using a standardised data extraction form, and independently checked for accuracy by a second reviewer. Information extracted will include details of the study's design and methodology, intervention and comparator tests, reference standard, baseline characteristics of participants, and outcome measures, including clinical outcome efficacy and any adverse events. Where there is incomplete information, if time allows, attempts will be made to contact authors with a request for further details. Discrepancies will be resolved by discussion, with involvement of a third reviewer if necessary.

3.5 Quality assessment strategy

The quality of diagnostic test accuracy studies will be assessed using the PROBAST (Prediction model Risk Of Bias ASsessment Tool) tool.^{39, 40} The quality of clinical effectiveness studies will be assessed based on their study design: randomised controlled trials will be assessed as per recommendations by the CRD³⁶ and the Cochrane Handbook for Systematic Reviews of Interventions⁴¹ and recorded using the Cochrane Risk of Bias Tool;⁴¹ non-randomised studies will be assessed using the Risk Of Bias In Non-randomized Studies-of Interventions (ROBINS-I) tool;⁴² and qualitative studies will be assessed using the Critical Appraisal Skills Programme (CASP) tool.⁴³ All quality appraisal assessments will be carried out by one reviewer and verified by another independently, with disagreements resolved by discussion or the involvement of a third reviewer, if necessary.

The results of the quality assessments might be used to inform sensitivity analyses to investigate the impact of study quality on the findings of the review through sensitivity analyses (where evidence permits).

3.6 Methods of analysis/synthesis

Details of results on clinical effectiveness and quality assessment for each included study will be presented in structured tables and as a narrative summary. Should clinically and methodologically homogenous studies be identified, data will be synthesised using appropriate meta-analytic techniques. Clinical, methodological and statistical heterogeneity will be investigated.

For test accuracy data, absolute numbers of true positive, false negative, false positive and true negative test results, as well as sensitivity and specificity values, with 95% confidence intervals will be presented for each study. Other measures of test accuracy data will be presented if reported.

3.6.1 Potential subgroup analyses

Subgroups of interest are:

- children versus adults with a diagnosis of Crohn's disease;
- newly diagnosed Crohn's disease versus established diagnosis of Crohn's disease;
- mild versus moderate-severe activity of disease;
- presence versus absence of fistulising or complex perianal disease.

3.6.2 Sensitivity analyses

Sensitivity analyses will be carried out including studies deemed to be high risk of bias (excluded from primary analyses).

4 REPORT METHODS FOR SYNTHESISING EVIDENCE OF COST EFFECTIVENESS

The economic evaluation will assess the cost-effectiveness of PredictSURE-IBDTM and Crohn's disease Prognosis Test (IBDX), compared with standard care in the NHS. The population to be included in the economic analysis consists of adults (aged 16 years and older) who have been newly diagnosed with moderate to severe Crohn's disease, and who have not been offered biologics under current standard care. A systematic literature review (SLR) of existing economic evaluations will be undertaken to inform the need, (and if necessary) the conceptualisation and development of a *de novo* economic model.

4.1 Identifying and systematically reviewing published costeffectiveness studies

A SLR will be undertaken to identify published full economic evaluations of the PredictSURE-IBD[™] and IBDX tools for the identification of those at high risk of developing a severe course of Crohn's disease, as well as economic evaluations of treatments for Crohn's disease. It is anticipated that searches conducted to retrieve records on the treatment of Crohn's disease will capture the relevant treatment strategies, including the "top-down" and "step-up" (standard and accelerated) approaches. In case such approaches are not identified through the SLR, a targeted search on "top-down" and "step-up" treatment strategies in the NHS will be conducted. A search filter to identify economic evaluations will be applied to the search strategies and the electronic databases will be searched from inception until the latest available version. The methodological quality of the full economic evaluations identified in the review will be assessed using the Drummond checklist.⁴⁴

The following databases will be searched for relevant studies:

- Ovid MEDLINE® In-Process & Other Non-Indexed Citations (Ovid);
- Ovid MEDLINE® ePub Ahead of Print (Ovid);
- Embase (Ovid);
- NHS Economic Evaluation Database (NHS EED) (Centre for Reviews and Dissemination, CRD);
- Cochrane Database of Systematic Reviews (CDSR) (Cochrane);
- Cochrane Central Database of Controlled Trials (CENTRAL) (Cochrane);
- Database of Abstracts of Reviews of Effects (DARE) (CRD);

• Health Technology Assessment Database (HTA) (CRD).

Separate searches will be carried out for supporting information on utility data. To identify cost and resource use evidence, the Evidence Assessment Group (EAG) will search the same sources identified for the economic evidence and treatment of Crohn's disease, together with NHS reference costs, the Unit Costs of Health and Social Care (Personal Social Services Research Unit [PSSRU]), the Electronic Marketing Information Tool (eMIT) and the British National Formulary (BNF). If the latter do not provide sufficient data to populate the economic model, a separate targeted search on costs and resource use will be conducted. Study selection and data extraction will be carried out as described in Sections 3.3 and 3.4, respectively.

The search strategies will combine terms capturing the interventions or comparators (Section Error! **Reference source not found.**) of interest and the target economic population aforementioned in Section 4. Health economic and quality of life search terms will be applied to capture the study designs of interest (cost-effectiveness, cost and quality of life, health state utility values [HSUVs]). Searches will be restricted to studies published in the English language; however, no restriction by setting or geographical location will be applied to the search strategy. If sufficient data are available from a UK setting, data from non-UK based studies will not be extracted. Moreover, if sufficient EQ-5D data are found during the searches for utility data, the EAG will restrict the data extraction to EQ-5D data.

In addition, clinical experts in the field will be contacted with a request for details of published and unpublished studies of which they may have knowledge. Furthermore, identified systematic reviews and submissions from companies will be searched for additional references.

Main findings from the studies identified from the SLR will be presented with a narrative synthesis and structured tables.

4.2 Development of a health economic model

Following the completion of the SLR, if a *de novo* economic model is deemed necessary, the EAG will develop the model in an appropriate software package (e.g., Microsoft® Excel) and using clinical expert opinion. The model will assess the cost-effectiveness of PredictSURE-IBDTM and IBDX, compared with standard care in the NHS, in adults (16 years and over) with a newly diagnosis of moderate to severe Crohn's disease, who have not been offered biologics under standard care. The cost-effectiveness analysis will assess the clinical and economic impact of early categorisation of patients' disease's risk (with the use of PredictSURE-IBDTM and IBDX) on disease management and clinical outcomes.

Model parameters (e.g., utilities, cost data) will be populated from the results of the economic and outcome searches and combined with unit costs from NHS reference costs and other relevant publications of UK health care costs as appropriate. The EAG will elicit expert opinion if published data are not available to inform all model parameters. All evidence will be evaluated according to the recommendations of the NICE Diagnostics Assessment Programme manual.³⁷

4.2.1 Model structure

The structure of the model will take into consideration any identified economic models in the area of Crohn's disease's treatment. The model structure will be dependent on the data that are identified through both the clinical and economic SLR.

Event pathways will be modelled to estimate long-term costs and benefits. The economic model will incorporate the pathways of care that individuals follow under standard practice in the UK NHS and for which credible evidence is available. The EAG will review previous economic models and seek expert clinical advice to help structure the diagnostic and care pathways.

The economic assessment will be undertaken from the perspective of the NHS and Personal Social Services. The model time horizon will be set to patient lifetime and both costs and benefits will be discounted at 3.5% per annum.

The output of the economic model will be incremental cost effectiveness ratios (ICERs), using qualityadjusted life-years (QALYs) as the measure of effectiveness. Various sensitivity analyses will be conducted to test the robustness of the model to changes in parameter assumptions and potentially also to alternative data sources. To assess the overall uncertainty in the model estimates, a probabilistic sensitivity analysis (PSA) will be conducted using appropriately sampled values for all relevant parameters in the model.

5 HANDLING INFORMATION FROM THE COMPANIES

All data submitted by the manufacturers/sponsors will be considered if received by the EAG no later than 15th July 2019. Data arriving after this date will not be considered. If the data meet the inclusion criteria for the review they will be extracted and quality assessed in accordance with the procedures outlined in this protocol.

Any 'commercial in confidence' data provided by a manufacturer and specified as such will be highlighted in <u>blue and underlined</u> in the assessment report (followed by company name in brackets). Any 'academic in confidence' data provided by manufacturers, and specified as such, will be highlighted in <u>yellow and underlined</u> in the assessment report. An executable model will be supplied, with any confidential data used in the cost effectiveness model replaced with dummy data.

6 COMPETING INTERESTS OF AUTHORS

None of the authors has any relevant competing interests to declare.

7 TIMETABLE/MILESTONES

Milestone	Date to be completed
Draft protocol	07/05/2019
Final protocol	03/06/2019
Progress report	30/08/2019
Draft assessment report	25/10/2019
Final assessment report	22/11/2019

8 **REFERENCES**

1. Abraham C, Cho JH. Inflammatory bowel disease. *N Engl J Med* 2009; **361**: 2066-78.

2. Hendrickson BA, Gokhale R, Cho JH. Clinical aspects and pathophysiology of inflammatory bowel disease. *Clin Microbiol Rev* 2002; **15**: 79-94.

3. Park JH, Peyrin-Biroulet L, Eisenhut M, Shin JI. IBD immunopathogenesis: A comprehensive review of inflammatory molecules. *Autoimmun Rev* 2017; **16**: 416-26.

4. Ananthakrishnan AN. Environmental triggers for inflammatory bowel disease. *Curr Gastroenterol Rep* 2013; **15**: 302.

5. Feagins LA, Iqbal R, Spechler SJ. Case-control study of factors that trigger inflammatory bowel disease flares. *World J Gastroenterol* 2014; **20**: 4329-34.

6. Iskandar HN, Ciorba MA. Biomarkers in inflammatory bowel disease: current practices and recent advances. *Transl Res* 2012; **159**: 313-25.

7. Johns Hopkins Medicine. Crohn's disease: introduction. 2001. Available from: https://www.hopkinsmedicine.org/gastroenterology_hepatology/_pdfs/small_large_intestine/crohns_d isease.pdf. Date accessed: 30 May 2019.

8.Crohn's and Colitis UK.Crohn's disease.2019.Available from:https://www.crohnsandcolitis.org.uk/about-inflammatory-bowel-disease/crohns-disease.Dateaccessed: 16 Apr 2019.

9. York Teaching Hospital. Crohn's disease. 2019. Available from: https://www.yorkhospitals.nhs.uk/our-services/a-z-of-services/inflammatory-bowel-disease-at-yorkhospital/crohns-disease/. Date accessed: 16 Apr 2019.

Crohn's and Colitis Foundation. Facts about inflammatory bowel diseases. 2019. Available
 <u>https://www.crohnscolitisfoundation.org/resources/facts-about-inflammatory.html</u>. Date
 accessed: 30 May 2019.

Lapidus A, Bernell O, Hellers G, Lofberg R. Clinical course of colorectal Crohn's disease: a
 35-year follow-up study of 507 patients. *Gastroenterology* 1998; 114: 1151-60.

12. National Institute for Health and Care Excellence. Crohn's disease: management. 2019. Available from: <u>https://www.nice.org.uk/guidance/ng129/chapter/Recommendations#inducing-remission-in-crohns-disease</u>. Date accessed: 30 May 2019.

13. Ghosh N, Premchand P. A UK cost of care model for inflammatory bowel disease. *Frontline Gastroenterol* 2015; **6**: 169-74.

14. National Institute for Health and Care Excellence. Crohn's disease: management. Clinical guideline. 2016. Available from: <u>https://www.nice.org.uk/guidance/cg152/resources/crohns-disease-management-pdf-35109627942085</u>. Date accessed: 16 Apr 2019.

15. Crohn's and Colitis Foundation. About the Epidemiology of IBD. 2019. Available from: https://www.crohnscolitisfoundation.org/resources/epidemiology.html. Date accessed: 17 Apr 2019.

16. Hovde O, Moum BA. Epidemiology and clinical course of Crohn's disease: results from observational studies. *World J Gastroenterol* 2012; **18**: 1723-31.

17. Garcia Rodriguez LA, Gonzalez-Perez A, Johansson S, Wallander MA. Risk factors for inflammatory bowel disease in the general population. *Aliment Pharmacol Ther* 2005; **22**: 309-15.

18. Rubin GP, Hungin AP, Kelly PJ, Ling J. Inflammatory bowel disease: epidemiology and management in an English general practice population. *Aliment Pharmacol Ther* 2000; **14**: 1553-9.

19. Maaser C, Sturm A, Vavricka SR, Kucharzik T, Fiorino G, Annese V, et al. ECCO-ESGAR Guideline for Diagnostic Assessment in IBD Part 1: Initial diagnosis, monitoring of known IBD, detection of complications. *J Crohns Colitis* 2019; **13**: 144-64.

20. Sturm A, Maaser C, Calabrese E, Annese V, Fiorino G, Kucharzik T, et al. ECCO-ESGAR Guideline for Diagnostic Assessment in IBD Part 2: IBD scores and general principles and technical aspects. *J Crohns Colitis* 2019; **13**: 273-84.

21. Peyrin-Biroulet L, Panes J, Sandborn WJ, Vermeire S, Danese S, Feagan BG, et al. Defining Disease Severity in Inflammatory Bowel Diseases: Current and Future Directions. *Clin Gastroenterol Hepatol* 2016; **14**: 348-54 e17.

22. Walmsley RS, Ayres RC, Pounder RE, Allan RN. A simple clinical colitis activity index. *Gut* 1998; **43**: 29-32.

23. Harvey RF, Bradshaw JM. A simple index of Crohn's-disease activity. *Lancet* 1980; 1: 514.

24. Vermeire S, Schreiber S, Sandborn WJ, Dubois C, Rutgeerts P. Correlation between the Crohn's disease activity and Harvey-Bradshaw indices in assessing Crohn's disease severity. *Clin Gastroenterol Hepatol* 2010; **8**: 357-63.

25. Gomollon F, Dignass A, Annese V, Tilg H, Van Assche G, Lindsay JO, et al. 3rd European Evidence-based Consensus on the Diagnosis and Management of Crohn's Disease 2016: Part 1: Diagnosis and Medical Management. *J Crohns Colitis* 2017; **11**: 3-25.

26. Regueiro M, Hashash JA. Overview of the medical management of mild (low risk) Crohn disease in adults. 2018. Available from: <u>https://www.uptodate.com/contents/overview-of-the-medical-management-of-mild-low-risk-crohn-disease-in-adults</u>. Date accessed: 18 Apr 2019.

27. Rogler G. Top-down or step-up treatment in Crohn's disease? *Dig Dis* 2013; **31**: 83-90.

28. Tsui JJ, Huynh HQ. Is top-down therapy a more effective alternative to conventional step-up therapy for Crohn's disease? *Ann Gastroenterol* 2018; **31**: 413-24.

29. Lee JC, Lyons PA, McKinney EF, Sowerby JM, Carr EJ, Bredin F, et al. Gene expression profiling of CD8+ T cells predicts prognosis in patients with Crohn disease and ulcerative colitis. *J Clin Invest* 2011; **121**: 4170-9.

30. McKinney EF, Lee JC, Jayne DR, Lyons PA, Smith KG. T-cell exhaustion, co-stimulation and clinical outcome in autoimmunity and infection. *Nature* 2015; **523**: 612-6.

31. McKinney EF, Lyons PA, Carr EJ, Hollis JL, Jayne DR, Willcocks LC, et al. A CD8+ T cell transcription signature predicts prognosis in autoimmune disease. *Nat Med* 2010; **16**: 586-91, 1p following 91.

32. Yi JS, Cox MA, Zajac AJ. T-cell exhaustion: characteristics, causes and conversion. *Immunology* 2010; **129**: 474-81.

33. Kamm F, Strauch U, Degenhardt F, Lopez R, Kunst C, Rogler G, et al. Serum anti-glycanantibodies in relatives of patients with inflammatory bowel disease. *PLoS ONE* 2018; **13**: e0194222. 34. Kaul A, Hutfless S, Liu L, Bayless TM, Marohn MR, Li X. Serum anti-glycan antibody biomarkers for inflammatory bowel disease diagnosis and progression: a systematic review and metaanalysis. *Inflamm Bowel Dis* 2012; **18**: 1872-84.

35. LabCorp. Crohn's Disease Prognostic Serological Marker Profile. A technical review. 2014. Available from: <u>https://www.labcorp.com/tests/related-documents/L6845</u>. Date accessed: 6 May 2019.

36. Centre for Reviews and Dissemination (CRD). CRD's guidance for undertaking reviews in healthcare. 2009. Available from: <u>https://www.york.ac.uk/media/crd/Systematic_Reviews.pdf</u>. Date accessed: 16 Apr 2019.

37. National Institute for Health and Care Excellence. Diagnostics Assessment Programme manual.
 2011. Available from: <u>https://www.nice.org.uk/Media/Default/About/what-we-do/NICE-guidance/NICE-diagnostics-guidance/Diagnostics-assessment-programme-manual.pdf</u>. Date accessed:
 16 Apr 2019.

38. Cochrane Methods. Screening and Diagnostic Tests. Handbook for DTA reviews. 2018. Available from: <u>https://methods.cochrane.org/sdt/handbook-dta-reviews</u>. Date accessed: 16 Apr 2019.

39. Moons KGM, Wolff RF, Riley RD, Whiting PF, Westwood M, Collins GS, et al. PROBAST: A Tool to Assess Risk of Bias and Applicability of Prediction Model Studies: Explanation and Elaboration. *Ann Intern Med* 2019; **170**: W1-W33.

40. Wolff RF, Moons KGM, Riley RD, Whiting PF, Westwood M, Collins GS, et al. PROBAST:
A Tool to Assess the Risk of Bias and Applicability of Prediction Model Studies. *Ann Intern Med* 2019;
170: 51-8.

41. Higgins J, Green S. Cochrane Handbook for Systematic Reviews of Interventions v.5.1.0. 2011. Available from: <u>http://handbook-5-1.cochrane.org/</u>. Date accessed: 18 Apr 2019.

42. Sterne JA, Hernan MA, Reeves BC, Savovic J, Berkman ND, Viswanathan M, et al. ROBINS-I: a tool for assessing risk of bias in non-randomised studies of interventions. *BMJ* 2016; **355**: i4919.

43. Programme. CAS. CASP Checklist: 10 questions to help you make sense of a Qualitative research. 2018. Available from: <u>https://casp-uk.net/wp-content/uploads/2018/01/CASP-Qualitative-Checklist.pdf</u>. Date accessed: 18 Apr 2019.

44. Drummond MF, Jefferson TO. Guidelines for authors and peer reviewers of economic submissions to the BMJ. *BMJ* 1996; **313**: 275-83.

9 APPENDICES

9.1 Appendix 1. Draft search strategy (MEDLINE)

9.1.1 Strategy relating to the prognostic tools

- 1 Crohn Disease/ (37,127)
- 2 Crohn*.mp (53,078)
- 3 ((Crohn\$ adj2 (disease or syndrome)) or regional enteritis).tw. (42,922)
- 4 Inflammatory bowel diseases/ (20,088)
- 5 IBD.mp. (22,385)
- 6 Inflammatory bowel disease*.mp. (48,006)
- 7 or/1-6 (84,427)
- 8 CD8-Positive T-Lymphocytes/ (34,740)
- 9 CD8+ T cells.mp. (34,151)
- 10 CD8 Antigens/ (8,663)
- 11 CD8 antigens.mp. (8,789)
- 12 CD8*.mp. (107,146)
- 13 CD 8*.mp. (1,182)
- 14 T-Lymphocytes, Regulatory/ (29,156)
- 15 Regulatory t cells.mp. (20,552)
- 16 (PredictSure or PredictImmune).mp. (0)
- 17 or/8-16 (139,264)
- 18 Antibodies/ (97,137)
- 19 antibod*.mp. (1,120,457)
- 20 glycan.mp. (15,592)

21	(antichitobiosic	de carbohydrate antibod* or ACCA or chitobioside).mp. (379)				
22	(antilaminaribi	oside carbohydrate antibod* or ALCA or laminaribioside).mp.	(291)			
23	(antimannobios	side carbohydrate antibod* or AMCA or mannobioside).mp.	(321)			
24	(anti-Saccharon	myces cerevisiae antibod* or ASCA or gASCA or mannan).mp.	(4,622)			
25	(anti-laminarin	carbohydrate antibod* or anti-L or laminarin).mp. (1,458)	I			
26	(neutrophil elas	stase degraded elastin or EL-NE).mp. (7)				
27	glycominds.mp	o. (4)				
28	(Crohn's disease prognosis test or IBDX).mp. (1)					
29	or/18-19 (1,120,457)					
30	or/20-28	(22,330)				
31	29 and 30	(4,874)				
32	7 and 17	(1,665)				
33	7 and 31	(421)				
34	32 or 33	(2,076)				
9.1.2	Strategy re	lating to interventions used to treat Crohn's dis	sease			
1	Crohn Disease/	(37,127)				
2	Crohn*.mp	(53,078)				
3	((Crohn\$ adj2	(disease or syndrome)) or regional enteritis).tw. (42,922)				

- 4 Inflammatory bowel diseases/ (20,088)
- 5 IBD.mp. (22,385)
- 6 Inflammatory bowel disease*.mp. (48,006)
- 7 or/1-6 (84,427)
- 8 prednisolone/ (32,042)

- 9 prednisone/ (38,447)
- 10 cortisone/ (19,547)
- 11 methylprednisolone/ (18,295)
- 12 hydrocortisone/ (70,241)
- 13 (corticosteroid or prednisolone or prednisone or methylprednisolone or hydrocortisone or budesonide).ti,ab. (120,262)
- 14 mesalamine/ (3,342)
- 15 sulfasalazine/ (4,014)
- 16 (mesalamine or sulfasalazine or "5-aminosalicylic*" or "5-aminosalicylate*" or "5-asa" or 5aminosalicylic* or 5aminosalicyclate* or 5asa or pentasa or mesalazine or mesalamine or asacol or sulfasalazine* or salazopyrin* or salazosulfapyridine* or asulfidine* or azulfadine* or azulfidine*).ti,ab. (7,653)
- 17 6-mercaptopurine/ (6,143)
- 18 azathioprine/ (14,342)
- 19 methotrexate/ (36,634)
- 20 (immunosuppressant or immunomodulator or mercaptopurine or methotrexate or amethopterin or Otrexup or Rasuvo or Rheumatrex or Trexall or Maxtrex or Nordimet or Zlatal or Methofill or Metoject or Jylamvo or azathioprine or Imuran or Azapress or thiopurine).ti,ab (67,727)
- (biologic or biologics or tumour necrosis factor alpha or TNF adj2 (inhibitor*)).ti,ab
 (4,152)
- 22 (infliximab or Remicade or Remsima or Inflectra or Zessly or Flixabi or adalimumab or Humira or Imraldi or Amgevita or Hulio or vedolizumab or Entyvio or ustekinumab or Stelara).ti,ab (15,541)
- 23 (top-down or top down or step-up or step up).ti,ab. (15,728)
- 24 or/8-23 (344,143)
- 25 7 and 24 (13,367)